Author's response to reviews

Title: Antipsychotic polypharmacy in a regional health service: a population-base study

Authors:

Miguel Bernardo (bernardo@clinic.ub.es)
Anna Coma (acomaf@catsalut.cat)
Cristina Ibañez (cibanezc@catsalut.cat)
Corinne Zara (czara@catsalut.cat)
Josep Maria Bari (jbari@clinic.ub.es)
Antoni Serrano-Blanco (aserrano@pssjd.org)

Version: 4 Date: 1 August 2011

Author's response to reviews: see over
1st August 2011

Editors-in-Chief, *Bio Med Central*

Dear Editor,

We send the last version of the manuscript entitled “Antipsychotic polypharmacy in a regional health service: a population-based study“ (Authors: Bernardo M, Coma A, Ibañez C, Zara C, Bari JM, Serrano-Blanco A.), which we have modified as suggested reviewers.

Please do not hesitate to contact us should you require any further information. I thank you in advance and look forward to hearing from you at your earliest convenience.

Yours sincerely,

Miguel Bernardo, MD, PhD